Author's response to reviews

Title: Antihyperglycemic and antihyperlipidemic activities of aqueous extract of Hericium erinaceus in experimental diabetic rats

Authors:

Bin Liang (liangtechb1@yahoo.com.cn)
Zhengdong Guo (guozhendong_cmu@yahoo.com.cn)
Fang Xie (xiefangcmu@yahoo.cn)
Ainong Zhao (zhaoainongcmu@yahoo.cn)

Version: 7 Date: 29 September 2013

Author's response to reviews:

Dear editors:

Thanks a lot for having reviewed our manuscript. The explanations regarding the reviewer’s suggestion are as follows:

Reviewer's report:

Although the authors have mentioned that two-way ANOVA was performed, there was no explanation about this thing in Table 3 and "Results" section. The significant differences of main effects (treatment and time) and their interaction (treatment × time) should be shown in Table 3 at least.

Thanks for your suggestion. I have revised the manuscript in the "results" according to your suggestion.

In the preliminary experiment, we did not find the difference of insulin levels in the experiment rats (untreated rats) during the experiment period. In two-way ANOVA analysis, we also found no interaction effects (treatment × time) on insulin levels.

Kind regards,

Liang